Author’s response to reviews

Title: Association of vitamin D receptor gene variants with polycystic ovary syndrome: A meta-analysis

Authors:

Xiao-Yuan Shi (15101246198@163.com)
Ai-Ping Huang (xly2008hi@163.com)
Duo-Wen Xie (329589337@qq.com)
Xiao-Long Yu (yxlzjh@sina.com)

Version: 2 Date: 19 Jan 2019

Author’s response to reviews:

We have re-phrased the textual overlap sections throughout the manuscript.

We have included a heading above the Declarations section.

We have amended the order of the Declarations.